Author's response to reviews

Title: NEWHINTS cluster randomised trial to evaluate the impact on neonatal mortality in rural Ghana of routine home visits to provide a package of essential newborn care interventions in the third trimester of pregnancy and the 1st week of life: Trial Protocol

Authors:

Betty R Kirkwood (betty.kirkwood@lshtm.ac.uk)
Alexander Manu (alex.manu@kintampo-hrc.org)
Charlotte Tawiah-Agyemang (charlotte.tawiah@kintampo-hrc.org)
Guus ten Asbroek (guus.tenasbroek@lshtm.ac.uk)
Thomas Gyan (thomas.gyan@kintampo-hrc.org)
Benedict Weobong (ben.weobong@kintampo-hrc.org)
R Eric Lewandowski (eric.lewandowski@gmail.com)
Seyi Soremekun (seyi.soremekun@lshtm.ac.uk)
Samuel Danso (danso.samuel@kintampo-hrc.org)
Catherine Pitt (catherine.pitt@lshtm.ac.uk)
Kara Hanson (kara.hanson@lshtm.ac.uk)
Seth Owusu-Agyei (seth.owusu-agyei@kintampo-hrc.org)
Zelee Hill (z.hill@ich.ucl.ac.uk)

Version: 2 Date: 22 February 2010

Author's response to reviews:

Proposal and ethics approval uploaded as requested.